\begin{thebibliography}{10}

\bibitem[Agarwal 90]{AgarwalEtAl90}
A.~Agarwal, B.~H. Lim, D.~Kranz, and J.~Kubiatowicz.
\newblock {APRIL}: A processor architecture for multiprocessing.
\newblock In {\em Proceedings of the 17th International Symposium on Computer
  Architecture}, pages 104--114. ACM and IEEE, 1990.

\bibitem[Alverson 90]{AlversonEtAl90}
Robert Alverson, David Callahan, Daniel Cummings, Brian Koblenz,
    Allan Porterfield, and Burton Smith.
\newblock The {Tera} computer system.
\newblock In {\em Proceedings of the 1990 International Conference on
  Supercomputing}, pages 1--6. ACM, 1990.

\bibitem[Ang 94]{AngArvChi94}
Boon~Seong Ang, Arvind, and Derek Chiou.
\newblock Star{T} the next generation: Integrating global caches and dataflow
  architecture.
\newblock Computation Structures Group Memo 354, M.I.T.
  Laboratory for Computer Science, February 1994.

\bibitem[Arvind 78]{ArvindGosPlo78}
Arvind, Kim P. Gostelow, and Wil Plouffe.
\newblock {\em An Asynchronous Programming Language and Computing Machine}.
\newblock Technical Report 114a, Department of Information and Computer Science,
  University of California, Irvine, December 1978.

\bibitem[Arvind 89]{ArvindNikPin89}
Arvind, Rishiyur S. Nikhil, and Keshav K. Pingali.
\newblock {I}-structures: Data structures for parallel computing.
\newblock {\em ACM TOPLAS}, 11(4):598--632, October 1989.

\bibitem[Cann 92]{Cann92}
David Cann.
\newblock Retire {F}ortran: {A} debate rekindled.
\newblock {\em CACM}, 35(8), August 1992.

\bibitem[Dennis 74]{Dennis74}
Jack B. Dennis.
\newblock First version of a data-flow procedure language.
\newblock In {\em Proceedings, Colloque sur la Programmation}, no.~19 in
  {\em Lecture Notes in Computer Science}, pages 362--376. Springer-Verlag, 1974.

\bibitem[Dennis/Gao 1994]{DennisGao94}
Jack~B. Dennis and Guang~R. Gao.
\newblock Multithreaded architectures: principles, projects, and issues.
\newblock In Robert~A. Ianucci, editor, {\em Advances in Multithreaded Computer
  Archtecture}. Kluwer, 1994.

\bibitem[Dennis/Gao 95]{DennisGao95}
Jack B. Dennis and Guang R. Gao.
\newblock Multiprocessor Implementation of Nondeterminate Computations
   in a Functional Programming Framework.
\newblock Computation Structures Group Memo 372, M.I.T.
  Laboratory for Computer Science, March 1995.

\bibitem[Hagersten 92]{HagerstenLanHar92}
Erik Hagersten, Anders Landin, and Seif Haridi.
\newblock {DDM}---A cache-only memory architecture.
\newblock {\em {IEEE} Computer}, 25(9):44--54, September 1992.

\bibitem[Hicks 93]{HicksEtAl93}
James Hicks, Derek Chiou, Boon~Seong Ang, and Arvind.
\newblock Performance studies of {I}d on the {M}onsoon dataflow system.
\newblock {\em Journal of Parallel and Distributed Computing}, 18(3):273--300,
  July 1993.

\bibitem[Hudak 91]{HudakEtAl91}
Paul {Hudak (editor)}, Simon {Peyton Jones (editor)}, Philip {Wadler (editor)},
  Brian Boutel, Jon Fairbairn, Joseph Fasel, Maria~M. Guzman, Kevin Hammond,
  John Hughes, Thomas Johnsson, Richard Kieburtz, Rishiyur Nikhil, Will
  Partain, and John Peterson.
\newblock Report on the programming language Haskell, a non-strict purely
  functional language (version 1.1).
\newblock Yale University, Department of Computer Science,
  August 1991.

\bibitem[Kuehn 88]{KuehnSmi88}
James T. Kuehn and Burton J. Smith.
\newblock The Horizon supercomputing system: Architecture and software.
\newblock In {\em Proceedings of Supercomputing '88}, pages 28--34.
   IEEE and ACM, November 1988.

\bibitem[Kuskin 94]{KuskinEtAl94}
Jeffrey Kuskin, David Ofelt, Mark Heinrich, John Heinlein, Richard Simoni,
  Kourosh Gharachorloo, John Chapin, David Nakahira, Joel Baxter, Mark
  Horowitz, Anoop Gupta, Mendel Rosenblum, and John Hennessy.
\newblock The {S}tanford {FLASH} multiprocessor.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture}, pages 302--313. IEEE and ACM, April 1994.

\bibitem[KSR 1992]{KSR92}
  Kendall Square Research.
\newblock KSR-1 Technical Summary. Waltham, MA, 1992.

\bibitem[Lamport 79]{Lamport79}
Leslie Lamport.
\newblock How to make a multiprocessor computer that correctly executes multiprocess
  programs.
\newblock {\em IEEE Transactions on Computers}, C-28:690--691, September 1979.

\bibitem[Lenoski 90]{LenoskiEtAl90}
Daniel Lenoski, Kkourosh Gharachorloo, James Laudon, Anoop Gupta, John Hennessy,
  Mark Horowitz, and Monica Lam.
\newblock Design of scalable shared-memory multiprocessors: The {DASH}
  approach.
\newblock In {\em COMPCON Spring '90: Digest of Papers}, pages 62--81.
  IEEE Computer Society, March 1990.

\bibitem[Lenoski 92]{LenoskiEtAl92}
Daniel Lenoski, James Laudon, Truman Joe, David Nakahira, Luis Stevens, Anoop
  Gupta, and John Hennessy.
\newblock The {DASH} prototype: Implementation and performance.
\newblock In {\em Proceedings of the 19th Annual International Symposium on
  Computer Architecture}, pages 92--103. IEEE and ACM, May 1992.

\bibitem[McGraw 85]{McGrawEtAl85}
J.~McGraw, S.~Skedzielewski, S.~Allan, R.~Oldehoeft, J.~Glauert, C.~Kirkham,
  B.~Noyce, and R.~Thomas.
\newblock {SISAL}: {S}treams and {I}teration in a {S}ingle {A}ssignment
  {L}anguage: {R}eference {M}anual {V}ersion 1.2.
\newblock Technical Report M-146, Rev. 1, Lawrence Livermore National
  Laboratory, 1985.

\bibitem[Papadopoulos 90]{PapadopoulosCul90}
Gregory M. Papadopoulos and David E. Culler.
\newblock Monsoon: An explicit token-store architecture.
\newblock In {\em Proceedings of the Seventeenth Annual International Symposium
  of Computer Architecture}, pages 82--91. IEEE and ACM, 1990.

\bibitem[Smith 81]{Smith81}
Burton~J. Smith.
\newblock Architecture and applications of the {HEP} multiprocessor computer
  system.
\newblock In {\em Proceedings of SPIE: Real-Time Signal Processing IV},
  volume 298, pages 241--248, 1981.

\end{thebibliography}

\section{Multithreaded Architecture Projects}
\label{related}

In this section, we breiefly discuss two related
multithreaded architecture projects.
These are the Monsoon project, which implements
a pogram execution model closely related to the one we have
presented; and the Tera computer and its forerunners, which also
address multiprocessor latency and synchronization issues.

\subsection{The Monsoon Project}

Several projects have designed and constructed multiprocessors to
execute nested function programs using dataflow principles.
Of these, the design that comes closest to fulfilling the
characteristics we have outlined is the Monsoon dataflow multiprocessor
~\cite{PapadopoulosCul90}.
The program execution model adopted in these machines derives from
the original {\em unravelling interpreter\/}~\cite{ArvindGosPlo78},
which has much in common with the graph/heap model of \cite{Dennis74}.
In Monsoon there are two forms of data memory, the frame store
and I-structure storage. These correspond, respectively, to
the data frames and heap storage of our implementation.
Monsoon uses dataflow graphs as an intermediate program form
for compiling. Each function application in Monsoon is executed entirely
by one processor, and all local variables are held in a stack
frame allocated in the local frame store.

Monsoon supports {\em I-structures}.
Incremental arrays are held in separate memory units that act
on messages sent by processors to request read and write operations.
Attempts to
read an undefined location in I-structure memory cause the
continuation of the processor activity to be entered in a defer queue
held in the I-structure memory. If the queue
has only one entry, it is stored in place of the expected
value; if more entries are needed, they are listed in an overflow area.
For more on I-structures, see~\cite{ArvindNikPin89}.
A write operation makes an entry in I-structiure memory defined, and
causes all waiting activities to be resumed with the element value.
Experiments with Monsoon have shown that defer queues
contain only a few entries during program execution, most often no more
than one~\cite{HicksEtAl93}.

There is no cache memory in Monsoon. Because the frame store is
local and is built of fast static RAM, there would be little benefit
in caching local variables. The implementation scheme
we have presented could be applied to the Monsoon I-structure memory.
It shows how caching of I-structure words may be done so as to
increase efficiency by reducing the number of network
transactions and decreasing the average latency of I-structure
accesses.

A successor project to Monsoon, the Start-NG
project~\cite{AngArvChi94}, plans to implement
a closely related program execution model using a
conventional superscalar microprocessor, the PowerPC chip.


\subsection{The HEP, Horizon and Tera}

The HEP, Horizon and Tera multiprocessor architectures
have considerable conceptual commonality with the designs
considered here. The processors are multithreaded and can
tolerate latency of memory accesses through the memory
network. Each processor has a large register set built of
fast static memory. The Tera commercial
supercomputer~\cite{AlversonEtAl90} is designed upon
the ideas and experiences from the Denelcor HEP
multiprocessor~\cite{Smith81} and the Horizon project at the
Supercomputing Research Center~\cite{KuehnSmi88}.
The Tera architecture includes four tag bits in each memory word
to support thread synchronization. These tags could be used
to support I-structure operations for incremental arrays, but
software support would be required to manage the queue of deferred
read operations and to schedule activities. To avoid excess
network traffic from repeated
attempts to complete a memory access blocked until a tag is set
by another process, an automatic retry mechanism is provided.

The Tera architecture does not include cache memory. Instead,
it depends on its large register set to reduce required main
memory bandwidth, and multithreading to ameliorate the
latency of main memory access.
Implementing caching as we have described above could improve
the efficiency of I-structure operations significantly.

The program execution model for these machines is a set of
independent processes running in a shared address space, and
the architecture does not provide any specific support for
function application as in Monsoon. Hence it appears that
support of modular nests of functions will be left to the
run-time software and compilers for high-level languages.



